Author's response to reviews

Title: A snapshot of the prevalence of physical activity amongst older, community dwelling Victorians: patterns across the young-old and old-old

Authors:

Jane Sims (j.sims@unimelb.edu.au)
Keith Hill (k.hill@nari.unimelb.edu.au)
Sandra Davidson (sdav@unimelb.edu.au)
Jane Gunn (j.gunn@unimelb.edu.au)
Nancy Huang (nancy.huang@heartfoundation.com.au)

Version: 4 Date: 16 February 2007

Author’s response to reviews:

Thank you for the acceptance of the revised m/s. I have now uploaded the document formatted as advised in your 'core pre-accept' message.